Author’s response to reviews

Title: Enhanced susceptibility to infections in a diabetic wound healing model

Authors:

Tobias Hirsch (Tobias.Hirsch@ruhr-uni-bochum.de)
Malte Spielmann (malte-spielmann@web.de)
Baraa Zuhaili (bzuhaili@partners.org)
Till Koehler (till.koehler@gmx.de)
Magdalena Fossum (mfossum@partners.org)
Hans-Ulrich Steinau (hans-ulrich.steinau@rub.de)
Feng Yao (fyao@rics.bwh.harvard.edu)
Lars Steinstraesser (lars.steinstrasser@rub.de)
Andrew B Onderdonk (aonderdonk@partners.org)
Elof Eriksson (eeriksson@partners.org)

Version: 4 Date: 18 February 2008

Author’s response to reviews:

1. Declaration of competing interests -> Competing interests

We have changed this paragraph according to the BioMed Central Editorial Production Team suggestion.

2. Affiliations: Each affiliation should be written in full, in the following format: department, institute, city, and country.

This may mean repeating institutional addresses if the authors are from different departments within an institution.

We have modified the affiliations according to the suggestions made by BMC.

3. References 15 and 16 have not been cited in the text and hence are missing from the extracted references list.

Please ensure that all the references are cited in the text at least once.

We have removed References 15 and 16 from the manuscript